Author's response to reviews

Title: Cortical injury in multiple sclerosis; the role of the immune system.

Authors:

Caroline A Walker (caroline.aldridge@yale.edu)
Anita J Huttner (anita.hutner@yale.edu)
Kevin C OConnor (kevin.oconnor@yale.edu)

Version: 3 Date: 3 December 2011

Author's response to reviews:

We have addressed the following requested changes:

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details

Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.

We have not made the following requested changes, because our manuscript is an invited review that does not conform to the suggested format:

Abstract: this abstract does not have the correct structure.

Structure: this manuscript does not have the correct structure.